\font\largerbold = cmbx10 scaled \magstep1
\font\largerrm = cmr10 scaled \magstep1
\font\largestbold = cmbx10 scaled \magstep2
\font\largerrm =cmr10 scaled \magstep1
\nopagenumbers
\magnification=\magstep2

\centerline{\largestbold Example} 
\bigskip

{\narrower
\noindent If you can read this then you should understand what is meant by 

\medskip
\centerline{\bf [PDF$|$PS$|$TeX]}
\medskip
\noindent so we can pleasantly write such expressions as 

$$7x=13\ {\rm mod}\ 17$$

\noindent and 

$$ 26^5=11881376$$

\noindent which is the number of 5 letter messages using the
(currently standard) alphabet. Maple computed this for me.

}
\smallskip
\centerline{\vrule height 1pt depth 0pt width 4truein}
\medskip
\parindent=4.2\parindent
\narrower
\noindent It will be nice when mathematical
typography is part of html.

\vfil\eject\end
